Minutes ~ RSC Weekly Meeting During RUN03 ~ April 11, 2003

 (1) Status of the AGS Commissioning [Leif]

     what has been accomplished...

     (a) no change in setup

     (b) source is working, but polarization was in the low 60s; a bit low compared to before the rf tube was replaced because Anatoli did not have time to tune it up until the last 24 hours.  Now it is in the high 60s/low 70s.  A bit more tuning might improve it to mid 70s.

     (c) source is more limited in terms of intensity - now it is a bit tough to get 1*10^11/pulse when it used to be that this edge was hit at 2*10^11

     what awaits...

     (a) finish setup so that AGS CNI can measure the polarization along the ramp [see George's comments below on this topic] and what else was discussed...

     (b) is the polarization stable at flattop?  or does it change on a

         long-time scale? 

     (c) is the measured, source polarization sign data [sign] being archived?  Waldo reported that this is not quite ready yet but he is pushing for it.

 (2) Status of the AGS CNI Polarimeter [George]

     what has been accomplished...

     (a) during the access on Wednesday 4/9, the silicon detectors were replaced due to increased noises (leakage current of 5-6 uA compared with 1-2 nA with fresh ones) arising from radiation exposure.  The new detectors are almost completely commissions and are quiet -- George showed that the alpha banana band was clearly visible in some of the low t-bins.  George thinks that the lower two t-bins are now sufficiently background free to be considered as good measurements.

     (b) replaced carbon targets.  Now, there are thinner targets (0.45 mm, 0.1 mm, and 0.07) in addition to the previous "thick" target of 0.6 mm.  The thinner targets should make it easier to use the polarimeter at higher intensities.

     (c) adjusted the preamplifier electronics to help solve some of the issues with high intensity running.

     (d) AND, MOST IMPORTANTLY, took a first stab at operating the CNI polarimeter along the ramp.  They see nice bananas and have the scaled GCC (gauss clock) in the system correctly.

     what remains to be done...

     (a) a complete measurement of the polarization along the AGS ramp.

     (b) measurements at a higher intensity; 'til now, they are restricted to 0.6x10^11 due to various background issues but would like to be at 1-2x10*11, the expected intensity of pulses which will be injected into RHIC.

 (3) Status of RHIC Commissioning [Haixin]

     what has been accomplished...

     (a) have ramped 110x110 beam with I=0.3x10^11/bunch and beta*=2m

     (b) have measured 35% polarization in blue at injection & flattop with I=0.5x10^11/bunch; measured zero for yellow 

     (c) have measured 40% polarization in yellow Friday morning at injection with 6x6; problems earlier were likely from the spin pattern

     (d) have tune feedback working for 6x6, now using it to develop beta*=1m tunes for PHENIX & STAR

      what awaits...

     (a) complete beta*=1m tuning

     (b) increase intensity/bunch

     (c) tune for luminosity at the experiments

     (d) measure polarization in yellow at flattop

     (e) commissioning of local polarimeters at STAR & PHENIX

     (f) commission the spin rotators [see minutes from last week for details] 

- turn on at store

         - measure polarization w/local polarimeters

     (g) commission the spin flipper [see discussion below for details]

     (h) downramp

     what additional discussions were there...

     (a) can we measure polarization along the ramp in RHIC? [Gerry]

         See below for details on this discussion.

 (4) Status of the RHIC CNI Polarimeters [Osamu]

     what has been accomplished...

(a) software – 

(*) all ready for one-button non-expert operation.

         (*) cut parameters for banana are measured and in place, but could benefit a bit from fine-tuning.

         (*) event-by-event ("AT") instead of histogram ("scaler") offline analysis code is ready.

(b) operations – 

(*) currently still using "scaler" mode (last year's setup) for measurements, but will want to switch to event-by-event mode.

         (*) has worked on 6x6 and 55x55 beams in RHIC, 110x110 failed due to a wrong file setting which has been fixed now.

         (*) QA code is almost ready.

     what additional discussions were there...

(a) telling the experiments known when a polarization measurement is underway? [Gerry]  

Last year, we used phone calls to the experiments.  This is a slow process which has inherent inefficiencies.  The proposal is to have a V102 "event" signal sent to the experiments with enough lead time so that the experiments can stop taking data.  We need to put together a concrete plan on this matter for discussion next week.

(b) measurement of polarization along the ramp? [Gerry]

If we can do it in the AGS, we can do it in RHIC.  RHIC is a bit easier because there are no spin flips along the ramp as in the AGS, so data can be blocked by time rather simply.  If we know something about the change of the asymmetry over the ramp, we might be better able to constrain the systematic error from the knowledge of the CNI analyzing power.  

     (c) measurements at injection in "AT" mode? [Osamu]

         Since the "AT" mode takes 5 to 6 minutes to readout the event-by-event data (there is definitely more than 5 min before finishing ramp), it would be practical to take the "AT" mode measurement only for later ring.  In this we can measure later ring, right after when we finish the measurement at the former ring.  Of course, we want to have the data on both ring at injection, thus it is preferred to have an alternating order of injection frequently. We will discuss this matter at this week's meeting.

 (5) Status of PHENIX [Doug]

     what has been accomplished...

     (a) PHENIX is ready to take data

     (b) from the 6x6 fill on Friday morning, PHENIX ZDC-based local 

         polarimeter showed measured LR asymmetries in neutrons which are in reasonable agreement with the results of the IP12 effort from last year.  The view is that PHENIX has a working local polarimeter; further statistics and software development over the weekend are planned for fast feedback to MCR.

     (c) CDEV data is percolating into our database

     what is planned...

     (a) continue to improve our DAQ performance to allow us to take even higher than last year rates.

     (b) continue to refine our calorimeter trigger so that we have sharper threshold curves.

     (c) continue to develop our muon triggers.

 (6) Status of STAR [Les]

     what has been accomplished...

     (a) collected some min-bias with STAR subsystems in the last couple of days.

     (b) the FPD electronics issues are largely solved and they are now

         completing the commissioning of the east detector.

     (c) the BBC + scaler board system has been checked using the "fastDAQ" mode.  There are still further checks underway, but things look good for having a BBC LR asymmetry from the Friday-Saturday overnight running if the beam conditions are good.

     what is planned...

     (a) complete the commissioning of the east FPD.

     (b) install the flash ADC readout for the west FPD on Wednesday.

     (c) measure BBC LR asymmetry using the scaler boards.

     (d) study relative luminosity, including the effect of multiple

         collisions.

 (7) Status of BRAHMS [Brendan]

     what has been accomplished...

     (a) BRAHMS is ready to take pp data and plans to start to collect

         the survey data as soon as 55x55 running overnight is stable.

     (b) the scaler boards (which are the same as those used at STAR) were tested in the lab and work

     what is planned...

     (a) start to take data overnight

     (b) install the scaler boards in the experiment and start to study the relative luminosity

 (8) Status of pp2pp [Stephan]

     what has been accomplished...

     (a) six of the eight detector packages are ready and tested.

     (b) the other two are almost ready and should be done by Monday.

     what is planned...

     (a) the detectors will be installed during the 2-day access on 

         Wednesday 4/16

     (b) development of hardware/software for readout, already underway, will be the focus then. 

 (9) Spin Pattern Proposal [Waldo]

     In coming up with a pattern, Waldo aimed to:

     (a) match the abort gaps at 4 & 10 o'clock

     (b) overlap so that bunch 1 & 11 (the funny bunches observed last year) collide with the abort gaps at STAR & PHENIX

     (c) keep some unpolarized bunches as per the RHIC polarimeter request

     (d) even number of ++/+-/-+/-- collisions at the spin experiments from which Waldo came up with a first stab at a pattern for 110 bunch running with 8 unpolarized bunches with:

                       PHOBOS    STAR    PHENIX/BRAHMS

                 0+

                 0-

                 00       8        4          4

                 +0

                 -0

                 ++      25       24         24

                 +-      26       24         24

                 -+      25       24         24

                 --      25       24         24

He also presented a plan with 3 and 6 unpolarized, but these do not have the same number of ++/+-/-+/-- collision pairs at the spin experiments. 

questions: 
(a) can we try random patterns? [Matthias]

To risky for this year because we want to be careful to make sure that we actually know the spin pattern.

                  (b) can we have 4 different patterns which we rotate through? [Gerry]

Probably this is nice a compromise between one fixed pattern and random patterns.




(c) can we avoid frequencies in the pattern? [Matthias]

Matthias points out that PHENIX has a "dead for 4 clocks" condition on its readout, so cycles of 5 could lead to a bias if the bunch intensities are not the same.  He would like to have no average Fourier component in the choice of 4 different patterns.

      plan: Gerry has agreed to the discussion of this matter during this week with the aim of having a conclusion by the Friday April 18th meeting.

(10) Commissioning the Spin Flipper [Mei]

     what has been accomplished...

     (a) partial success of last year, even though there was minimal

         commissioning time, indicate that this device has great promise for this year.  Since it reduces the relative luminosity systematic error significantly, we feel that it should get some time to see if it can be made operational for this year, especially since it also provides useful information about spin in the machine.

     (b) the hardware is completely ready.

     (c) the software for operating it in non-expert mode is almost ready; the experiments need to learn how to use the V102 event signals.

     what is needed...

     (a) measurement of the flip efficiency -vs- flipper frequency rate

     (b) measurement of the flip efficiency -vs- flipper strength at fixed frequency 

     (c) measurement of the flip efficiency -vs- spin tune of the snake each of these measurements take 3 or so hours.  Mei prefers to have chunks of 3+ hour here and there instead of one continuous period since she can then learn from a measurement prior to doing the next one.

     DECISION: It was decided that the spin flipper commissioning should be undertaken before the Wednesday access.  To avoid delaying any other machine commissioning efforts (i.e. the spin rotator commissioning), the experiments agreed to give up a night of running for this effort.  Haixin and Mei will decide how best to proceed.

(11) PHENIX BBC asymmetry with a tag of neutron/gamma in the ZDC [Kiyoshi]

     From further analysis of the PHENIX Run-2 data, Kiyoshi sees a significant BBC LR asymmetry for data in which there is one hit in the BBC and a hit in the ZDC.  The AN power looks like it is about 3%, independent of whether the pulse height of the BBC is below, in, or above the minimally-ionizing peak.  We hope that these data will provide an additional local polarimeter measurement for PHENIX.  It may also be a nice thing for STAR to look at too since it might have a bit larger of a figure of merit than their BBC asymmetry and would only require the additional of ZDC signals to their scaler module.  

